Author's response to reviews

Title: Disparities Exist Between National Food Group Recommendations and the Dietary Intakes of Women

Authors:

Michelle L Blumfield (Michelle.Blumfield@newcastle.edu.au)
Alexis J Hure (Alexis.Hure@newcastle.edu.au)
Lesley K MacDonald-Wicks (Lesley.Wicks@newcastle.edu.au)
Amanda J Patterson (Amanda.Patterson@newcastle.edu.au)
Roger Smith (Roger.Smith@newcastle.edu.au)
Clare E Collins (Clare.Collins@newcastle.edu.au)

Version: 5 Date: 4 August 2011

Author's response to reviews: see over
4th August, 2011

Dear Dr Will,

Thank you for your email. We have cited reference 11 in the manuscript as requested.

Yours sincerely,

[Signature]

Professor Clare Collins, PhD, advAPD, FDAA
NHMRC Research Fellow
On behalf of all authors